Author’s response to reviews

Title: Intraoperative TOE guided management of newly diagnosed severe tricuspid regurgitation and pulmonary hypertension during orthotopic liver transplantation: a case report demonstrating the importance of reversibility as a favorable prognostic factor.

Authors:

Brett Pearce (brett.pearce@austin.org.au)
Raymond Hu (raymond.hu@austin.org.au)
Fiona Desmond (fiona.desmond@austin.org.au)
Daniel Banyasz (daniel.banyasz@austin.org.au)
Robert Jones (robert.jones@austin.org.au)
Chong Tan (chong.tan@austin.org.au)

Version: 2 Date: 14 May 2019

Author’s response to reviews:

Dear editor,

Thank you for once again giving us the opportunity to re-submit our case report.

We have emailed through the change of authorship form and made the minor changes suggested (changed the words Case Summary to case presentation in the abstract). A single clean version has been uploaded to the system. All declarations have been checked.

Should you require anything further please do not hesitate to contact us.

Kind regards

Dr Brett Pearce